Author's response to reviews

Title: Hepatitis B Vaccinations Among Koreans: Results from 2005 Korea National Cancer Screening Survey

Authors:

Hee-Soon Juon (hjuon@jhsph.edu)
Kui Son Choi (kschoi@ncc.re.kr)
Eun-Cheol Park (ecpark@ncc.re.kr)
Min-Son Kwak (serene@ncc.re.kr)
Sunmin Lee (sunmin@umd.edu)

Version: 5 Date: 12 November 2009

Author's response to reviews:

November 12, 2009
Melissa Norton, MD
Editor-in-Chief, BMC Infectious Disease
The BioMed Central Editorial Team

RE: Manuscript – 1733590869257367: Hepatitis B Vaccinations Among Koreans: Results from 2005 Korea National Cancer Screening Survey

Dear Dr. Norton

My colleagues and I are pleased to hear that our manuscript was accepted. We have made the changes to the format of our manuscript as suggested.

Thank you in advance for your consideration.

Sincerely,
Kui Son Choi, PhD.
National Cancer Control Institute,
National Cancer Center,
111, Jungbalsan-ro, Ilsandong-gu, Goyang-si, Gyeonggi-do, 410-769, Korea
Tel: +82-31-920-2192; Fax: +82-31-920-2189; E-mail: kschoi@ncc.re.kr